Author’s response to reviews

Title: An increase of cereal intake as an approach to weight reduction in children is effective only when accompanied by nutrition education: a randomized controlled trial

Authors:

Jorge L Rosado (jlrosado@avantel.net)
Maria del R Arellano (roare2001@hotmail.com)
Karina Montemayor (carim7@yahoo.com.mx)
Olga P Garcia (opgarcia@yahoo.com.mx)
Maria del C Caamaño (mccaamano@hotmail.com)

Version: 4 Date: 19 August 2008

Author’s response to reviews:

August 18th, 2008

The Nutrition Journal Editorial Team

Nutrition Journal

Re: MS: 3708766121934397

Attached please find a revised version of the paper entitled “An increase of cereal intake as an approach to weight reduction in children is effective only when accompanied by nutrition education: a randomized controlled trial”.

We reviewed the paper and made the format changes suggested by the BioMed Central Editorial Production Team.

We authorize the editor to make any other formatting change or typographical errors that we have missed.

Sincerely,

Jorge L. Rosado, PhD.
Senior Scientist
School of Natural Sciences
Universidad Autónoma de Querétaro